Author's response to reviews

Title: Vaccination coverage in children can be estimated from health insurance data

Authors:

Helen Kalies (kalies@uni-muenchen.de)  
Rebekka Redel (rebekka.granda@kvb.de)  
Rudolf Varga (rudof.varga@aok.bayerns.de)  
Martin Tauscher (martin.tauscher@kvb.de)  
Rudiger von Kries (Prof.von.Kries@gmx.de)

Version: 5 Date: 23 February 2008

Author's response to reviews:

Dear Editor and BioMed Central Editorial Production Team,

Thank you for accepting the last version of our manuscript. You asked us to make some changes to the format of our manuscript so that online production process of our manuscript will be straightforward.

We considered all your points, in detail:

- we removed the authors' qualifications from the authors list on the title page.
- we deleted the running head, key words and the word count.
- we now used sentence case for all headings
- we renamed Introduction in `Background'
- we wrote `The authors declare that they have no competing interests' in competing interests
- we deleted the title at the Figure title
- we cropped the figure

and we corrected two typos.

We now are looking forward to the publication of our manuscript,

best regards,

Helen Kalies
in the name of all authors